Hall D Controls Meeting 02-May-2019

Date: 05/02/19 Time: 10:00am

<u>Attendees</u>: Peter Bonneau, Pablo Campero, Hovanes Egiyan, Brian Eng, Amanda Hoebel, Tyler Lemon, Nick Sandoval, Scot Spiegel, Tim Whitlatch

1. General:

- 1.1. Next week, DSG will work on PXI.
- 1.2. UPS failed during power outage on April 22.
 - 1.2.1. All servers on UPS were rebooted.
 - 1.2.2. Some PLCs did not reconnect to network when power came back on.
- 1.3. Cables will be installed for VME crate remote reset.
- 2. Solenoid status:
 - 2.1. Solenoid warmed to 80K for refrigerator maintenance.
 - 2.2. Vacuum alarms on PLC disabled for magnet warm-up.
- 3. ComCal:
 - 3.1. New air-cooled chiller ordered to replace old chiller. 3.1.1.Due to arrive in June.
 - 3.2. Tentative plan to install detector in the fall.
- 4. DIRC controls:
 - 4.1. Liquid level sensors tested for optical box.
 - 4.2. Window scheduled to arrive in mid-May. 4.2.1.Installation scheduled for June.
 - 4.3. Need temperature readout from FPGA board to be put in EPICS.